Contents lists available at ScienceDirect

Journal of Health Economics

journal homepage: www.elsevier.com/locate/econbase

Maternity ward crowding, procedure use, and child health $\!\!\!\!\!^{\updownarrow}$

Jonas Maibom^{a,*}, Hans H. Sievertsen^b, Marianne Simonsen^c, Miriam Wüst^d

^a Aarhus University and IZA, Denmark

^b University of Bristol, VIVE, and IZA, United Kingdom

^c Aarhus University, Denmark

 $^{\rm d}$ University of Copenhagen, CEBI, VIVE, and IZA, Denmark

ARTICLE INFO

Article history: Received 21 November 2019 Received in revised form 27 August 2020 Accepted 1 September 2020 Available online 22 November 2020

ABSTRACT

This paper studies the impact of day-to-day variation in maternity ward crowding on medical procedure use and the health of infants and mothers. Exploiting data on the universe of Danish admissions to maternity wards in the years 2000–2014, we first document substantial day-to-day variation in admissions. Exploiting residual variation in crowding, we find that maternity wards change the provision of medical procedures and care on crowded days relative to less crowded days, and they do so in ways that alleviate their workload. We find very small and precisely estimated effects of crowding on child and maternal health. Thus our results suggest that, for the majority of uncomplicated births, maternity wards in Denmark can cope with the observed inside-ward variation in daily admissions without detectable health risks.

© 2020 Elsevier B.V. All rights reserved.

1. Introduction

Childbirth is one of the most common reasons for hospitalizations in developed health care systems. Across countries and over the past decades, the typical hospital birth has changed: We have witnessed an increase in the use of medical procedures, such as Caesarean sections (CS) and inductions, not only for at-risk populations but also for a more general and healthy population of births. While an

Corresponding author.

h.h.sievertsen@bristol.ac.uk (H.H. Sievertsen), msimonsen@econ.au.dk (M. Simonsen), miriam.w@econ.ku.dk (M. Wüst).

https://doi.org/10.1016/j.jhealeco.2020.102399 0167-6296/© 2020 Elsevier B.V. All rights reserved. existing literature has documented large benefits of additional medical care and specialized medical technologies for at-risk births in the short- and longer-run (Almond et al., 2010; Almond and Doyle, 2011; Bharadwaj et al., 2013; Daysal et al., 2015; Jensen and Wüst, 2015), we lack causal evidence on the impact of the level and variability of medical care that is informative for the majority of ex ante uncomplicated births. This paper starts filling this gap by studying the impact of one factor that may impact both the allocation of care and patient health, namely naturally occurring variation in the number of admissions to maternity wards or temporary maternity ward crowding. Thus we ask: What is the causal impact of being admitted for birth on a busy day rather than a less busy day on both procedure use and on child and maternal health?

Crowding breaks the link between underlying patient characteristics and assigned health care treatment because of the nature of maternity ward admissions: While wards can anticipate some of the variation in admissions (Macfarlane, 1978; Allin et al., 2015; Johnson and Rehavi, 2016), they cannot perfectly predict it. Maternity



ECON



[☆] Wüst acknowledges financial support from the Tryg Foundation and the Danish Council for Independent Research, grant 8019-00051B. Simonsen acknowledges financial support from Spar Nord Fonden. We thank Simon Burgess, Helena Skyt Nielsen, Stephanie von Hinke Kessler Scholder, Christine Valente, seminar participants at the University of Copenhagen, the University of Bergen, Lund University, and the ESPE (2018), EALE (2018) and ASSA (2019) conferences for helpful comments. We thank Ida Lykke Kristiansen for excellent research assistance.

E-mail addresses: maibom@econ.au.dk (J. Maibom),

wards operate in a similar fashion as emergency rooms, with admissions that are hard to schedule for (expected) uncomplicated birth. Thus our empirical strategy exploits variation in daily maternity ward admissions conditional on ward, year, season, day of week and ward times year fixed effects. Controlling for these factors allows us to study the impact of the residual variation in crowding beyond the expected admission load on a given day of admission for otherwise similar patients with a spontaneous birth.¹

To examine the impact of crowding on procedure use and health, we exploit data on all admissions to Danish maternity wards in the years 2000–2014. In our data, we observe large day-to-day variation in admissions of well above 100 percent of mean admissions, and deviations of above 200 percent are not rare. Furthermore, the withinward (over time) variation in crowding is similar to the level of variation we observe across maternity wards, i.e. in the number of daily admissions across small and large wards.

We first show that maternity wards change their care provision due to unexpected fluctuations in the number of daily admissions. This analysis is informative both in its own right but also for understanding potential consequences for patient health: Capacity or lack hereof may be among non-medical factors that impact procedure use. We find that maternity wards use fewer procedures for births admitted on crowded days relative to less crowded days.² Specifically, we find small but significant decreases in the probability of experiencing a medically induced birth. These findings may indicate that maternity ward staff attempt to alleviate pressure by avoiding to speed up deliveries (and postpone inductions). This finding is interesting in the light of the steadily increasing induction rates in many developed health care systems and accompanying debates on the benefits of the procedure in marginal cases. Moreover, in supplementary analyses on a subset of our data, we find that maternity wards adjust the timing of admission on a given day during busy periods: Mothers admitted on crowded days give birth faster, i.e. they are likely admitted at later stages of their birth. Likely related to the (on average) shorter stays at the maternity ward for women in labor, we find that those admitted on crowded days are less likely to receive pain relief, measured as the probability of having an epidural. Finally, while we see different types of adjustments in the wards with respect to procedures used and their timing, we do not detect effects on waiting times for pain relief in the form of epidurals (for those receiving an epidural), the timely initiation of skin-to-skin contact between the mother and the infant, and mothers' probability of having an emergency CS (our closest measures of patient experience).

Next, we examine the health consequences of temporary crowding on the day of the mother's admission for birth outcomes and child health in the first two years of life. We show that while crowding induces changes to the provision of care, it has only small effects on child and mother health in our sample. These health outcomes include infant well-being as measured by the probability of having a low APGAR score (a measure for infant Appearance, Pulse, Grimace, Activity, and Respiration) at 5 min; the probability of experiencing birth and post-birth complications for mothers; children's mortality, post-birth hospital stay, neonatal care admission, and readmissions and contacts to general practitioners (GPs) during the first two years of their life. Among these measures we specifically consider outcomes that relate to staff intensity (severe lacerations during birth) and severity (post-birth maternal complications that require additional medical procedures or operations). For maternal complications at and shortly after birth we find small but significant decreases due to more crowding-a result that likely relates to the decreased use of procedures such as inductions and pain relief on crowded days.

Our results for both health outcomes and procedure use are similar across alternative measures of temporary crowding such as an absolute measure based on the number of daily admissions and relative measures defined within a given ward. Furthermore, we do not find evidence of threshold effects, i.e. we do not find that our main results based on a continuous crowding measure and a linear regression model mask heterogeneity by the level of crowding (in the range observed in our data). Finally, we find limited evidence for heterogeneity in the effects across characteristics of the birth or hospital. Our findings suggest that the effect of crowding on the probability of experiencing an induction is strongest for first-born children, children born at a gestational age above 41 weeks, and children admitted to the largest maternity wards. Given that pregnancies at term or after are the main target group for inductions, our finding of stronger effects of crowding on inductions for beyond-term births and the absence of detectable health effects may indicate that maternity wards provide too many inductions on less busy days.

Our work contributes to an established literature on the impact of crowding for health care provision and health outcomes.³ In this literature, studies vary in (i) the health care settings studied (and, as a result, the patient pool and health care organization considered), (ii) their measure of crowding (number of admissions, staff-patient ratios) and (iii) their approach towards handling unobserved factors that may impact both, crowding and patient health.

A number of studies has shown negative associations between crowding and patient health outcomes across settings (see, for example, Aiken et al., 2002; Morley et al., 2018).⁴ Building on these correlational findings

¹ Importantly, as the patient-staff ratio is likely endogenous to the riskprofile of the admitted women, and as we cannot observe the actual patient-staff ratio or the staff assigned to a specific birth, we focus on the reduced form impact of being admitted on a busy vs. a less busy day. We detail these points in Section 2.

² While we account for scheduled CS in our measure of crowding, we do not assess the impact of crowding on scheduled CS as those are scheduled well in advance and taking many factors into account. We detail our identification strategy in Section 3.

³ Related to this literature, a broader question in health economics concerns the importance of variations in health care resources (Chandra et al., 2011; Skinner, 2011) and ways to identify the health impacts of access to these resources (see, for example, Doyle, 2011; Doyle et al., 2015).

⁴ For example, reviewing evidence from studies on crowding at the emergency department, Morley et al. (2018) conclude that crowding and patient health are (negatively) related. Using hospital level information

that predominantly use across-hospital variation, the literature on the impact of crowding has moved towards causally-identified estimates. These causal studies fall in two broad strands that either exploit legislative changes that permanently alter staffing levels (Cook et al., 2012; Matsudaira, 2014), or use sudden (within) hospital shocks that temporarily change the staff-patient ratio to identify the impact of crowding (Evans and Kim, 2006; Facchini, 2017; Woodworth, 2020). Our work is closely related to the second strand of studies, which-across different health care settings-has typically found smaller (if any) impact of changes in the patient-staff ratio or crowding shocks on health outcomes: Exploiting unexpected surges in hospital admissions that result in changes in the staff-patient ratio, Evans and Kim (2006) find only limited evidence for impacts of large shocks to hospital admissions on the length of the hospital stay and the risk of a subsequent readmission. Focusing on emergency room care, on the other hand, Woodworth (2020) uses sudden crowding shocks in established emergency departments due to the opening of new emergency departments to show that an alleviation of emergency department patient volume does matter for patient health: lower volumes lead to significantly lower patient mortality.⁵ We complement this existing work on temporary crowding by providing evidence from the maternity ward - an until now under-studied setting. Moreover, our estimates for the health effects of crowding focus on an (on average) more healthy patient group than considered in much of the existing work: mothers, who give birth spontaneously. Finally, we study the policyrelevant margin of day-to-day variation in crowding. This analysis distinguishes our paper from studies that consider the impact of larger crowding shocks (such as closures or openings of new wards).

Besides providing estimates for the health impacts of crowding, we contribute novel evidence on crowding and the allocation of care and procedures in the maternity ward. Evidence on the ways in which crowding influences treatment decisions is scarce but instrumental for our understanding of drivers of medical care allocation. Existing research shows that crowding impacts decisions on the allocation of care: Harris et al. (2019) show that higher work-load in primary care settings leads to fewer procedures, fewer diagnoses, and more referrals. Within the area of neonatal care, Freedman (2016) exploits variation in empty beds at hospital neonatal inten-

sive care units (NICUs) to document that short-term excess capacity increases NICU utilization among relatively lowrisk infants but does not impact usage among at-risk infants. This finding suggests that the greater availability of resources affects the provision of care, especially for infants where the treatment decision is highly dependent on the discretion of medical professionals. Facchini (2017) uses data from one Italian maternity ward to document that crowding affects decisions about care allocation and procedure use. Specifically, women admitted for birth at times with a low midwife/patient ratio are more likely to receive an unscheduled CS than women admitted at times with better coverage. He also studies health outcomes and-in line with other causal studies on crowding-finds no significant effects on measures such as skin-to-skin contact, breastfeeding, APGAR<9, or NICU admissions but large standard errors preclude a conclusion of no important health effects.⁶

While we do not observe the patient-staff ratio (as in Facchini (2017)), our paper contributes new evidence on the causal impact of crowding on a broader set of medical treatments assigned to births (other than CS), and we are not constrained to one maternity ward and resulting power issues. Thus we extend on a closely related study by Marks and Choi (2019), who study the impact of temporary crowding at the maternity ward on the health of at-risk infants and health care spending in the U.S. They find that at-risk infants born on "slow days" receive more costly care (measured as overall hospital spending)-but they find limited mortality benefits of this additional spending. By studying procedure use and health outcomes in maternity wards in Denmark, we highlight the importance of accumulating empirical evidence on central policy questions across different settings: First, we open the black box of health care spending by providing evidence on the impact of crowding on specific procedures and treatments. Second, while Marks and Choi (2019) focus on the impact of crowding on the 10 percent at-risk infants in their data, we study the majority of non at-risk infants. We document changes in care and procedures provided for this group, while Marks and Choi (2019) do not find that crowding predicts spending for low-risk births in the U.S. This finding implies that also care for low-risk infants is susceptible to the impact of crowding.⁷ Third, by considering a different margin of the patient health distribution, we confirm that, although

about ambulance diversion hours, Sun et al. (2013) find that emergency department crowding increases mortality and hospital stays. Sprivulis et al. (2006) show that crowding at the emergency room is associated with increased mortality. Closely related to our work, Snowden et al. (2017) use variation in crowding around actual delivery days (i.e. not on the day of admission) and compare busy days to less busy days at Californian maternity wards. They find that a busy day is related to significantly worse infant health outcomes (baby well-being as measured by a value on the APGAR score below 7, neonatal intensive care admissions, and length of post-birth hospital stay).

⁵ In a related paper, crowding at emergency departments has also been shown to increase the costs of care: Exploiting variation induced by assignment of patients to different nurses, Woodworth and Holmes (2019) show that prolonged waiting time for patients with serious conditions increases hospital costs to care for the patient.

⁶ Related to these findings, Avdic et al. (2018) provide indirect evidence on the impact of crowding in their study of maternity ward mergers. Those mergers resulted in the closure of smaller wards in Sweden. While they show that mergers do not have negative health effects for newborns, they find small negative impacts of hospital mergers on maternal health, as measured by a summary measure of maternal trauma. They show that a contributing factor for their result may be crowding that decreases the use of a central medical procedure: Caesarean Sections. As discussed above, in our setting, we do not find that crowding results in changes in the probability of having an emergency CS, which may explain, why we do not find strong post-natal health effects on mothers in our study.

⁷ We also argue that we only can isolate exogenous variation in crowding for the non at-risk group given that hospitals typically schedule CS for at-risk pregnancies and this practice violates our identifying assumption. Furthermore, we abstain from instrumental variable strategies for the impact of crowding (applied in Marks and Choi (2019)), because we argue that crowding may impact infants through many channels-not all

crowding impacts procedure use also in this patient group, health effects are limited.

In sum, our paper contributes causally-identified evidence on crowding and its limited impact on patient health for a general and (on average) healthy population of patients in a so far understudied setting: the maternity ward. Moreover, we show that crowding impacts procedure use and thus we contribute new evidence on drivers of medical care allocation. Important for health care policy, our findings on the impact of crowding in Danish maternity wards suggest that wards react to temporary crowding by adjusting medical care assignment across several margins, among them the exact timing of admission during the day and the use of medical procedures. While our findings are not informative about the optimal absolute level of care in the maternity ward, we conclude that maternity wards in the Danish setting can balance natural variation in admissions for a general population of spontaneous births without negative health effects in the outcomes that we can study.

The paper proceeds as follows: Section 2 provides information on the institutional background and our data. Section 3 presents our empirical strategy and discusses the identifying assumptions. Section 4 presents our main results and examines their robustness. Finally, Section 5 concludes.

2. Background and data

2.1. Danish health care services during pregnancy and at birth

Health care around birth is publicly provided and free of charge for all residents in Denmark. Prior to birth, GPs and midwives provide prenatal care. Home births account for around two percent of all births in Denmark, and there are no private maternity wards. Hospital assignment occurs in the first trimester of pregnancy and, by default, women give birth at the hospital of their catchment area. However, in the event of excess capacity, pregnant women can choose freely among all other public hospitals. Anticipated highrisk births may be assigned to specialized hospitals.

Hospital births for (expected) uncomplicated births in Denmark are performed in the maternity ward and are as a default midwife-assisted. In our data period, we document that women on average spend around 10 h in the maternity ward (an average that masks large heterogeneity). During the course of labor, midwives (in cooperation with the prospective parents) determine treatment and procedure choices. Only in case of complications the midwife consults a physician, and all maternity wards have specialist physicians as backup for the case of complications during birth. At-risk births (that are performed as scheduled CS) do not enter the maternity ward on the described terms but are treated by a pre-allocated team of staff members including obstetricians, anesthesiologists, nurses, and midwives. After uncomplicated hospital births, mothers and infants are either admitted to a separate hospital ward for continued care and observation or discharged on the same day. Postnatal hospital care involves guidance and care by hospital-based nurses (to help women to establish infant feeding), as well as health checks and screenings. Outpatient care after birth for higher parity mothers (and increasingly also first-time mothers) is common in Denmark. Mothers return to the hospital for health checks and screenings and receive early postnatal home visits by trained home visitors in the Danish home visiting program. In our sample, 36 percent of mothers are discharged on the day of birth.⁸

2.2. Data

We use data on all births (including stillbirths and multiple births) during 2000–2014 from the Danish Medical Birth Register.⁹ From the raw data, we omit births that take place at home, resulting in 945,072 relevant births.¹⁰

We link these birth records to a number of other administrative data sources providing us with a set of individual level control variables for a range of pregnancy and parental characteristics, including an indicator of pregnancy complications, indicators for parental educational status (in education, completed higher education, completed university), parental non-western origin, parental disability pension status and parental disposable income (2010 level).¹¹

2.2.1. Treatment variable and analysis sample

To construct our main treatment variable of crowding on the day of admission for birth, we link the birth records to the Danish Inpatient Register. Not all mothers are admitted to hospital on the day they give birth. To assign the relevant day of admission to each birth, for each mother we identify the hospital admission date that is closest to the date of birth of the relevant child and that is not separated from this date by a hospital discharge date. For 98 percent of mothers in our sample, the birth admission date to a maternity ward lies during the 3 days up to the birth of the child; for 94 percent, the relevant admission is on the day before or on the day of their child's birth. We use the full sample of hospital births to construct our treatment variables measuring crowding on the day of the birth-related hospital admission for each mother.¹² Thus we include scheduled CS and multiple births, the latter with the num-

of which are observed. For further details on our identification strategy, see Section 3.

⁸ In our results section, we examine whether crowding impacts the probability of a same-day discharge for mothers.

⁹ As stillbirth we define births that have completed at least 170 days of gestation and that have a death date on the day of birth. We exclude registered births with shorter gestation.

¹⁰ Mothers have to make the decision to give birth in an assisted home birth already during the pregnancy.

¹¹ All characteristics are measured two calendar years prior to the relevant child's birth. For missing values, we set the value to zero and include an indicator variable for missing values for each of the covariates.

¹² We subtract the focal mother from the number of admissions on her admission day.

ber of children in the birth.¹³ We construct two measures of maternity ward crowding: (i) the absolute (leave out) number of admissions to each maternity ward on any admission day and (ii) the percentile rank (between 0 and 1) of the given admission day (within hospital, season (quarter) and year cells).¹⁴

Having computed our measures of crowding on our full sample of hospital births, we constrain our estimation data in the following three ways: First, given that our analysis relies on an extensive set of fixed effects (i.e. compare outcomes in cells defined by year, ward, maternity ward \times year, season, and day of the week cells), we drop births in small year × ward cells of less than 700 births (3 percent of the sample). Second, we omit scheduled CS (around 9 percent of our sample) and multiple births (in our sample, more than half of the multiple births are delivered through a scheduled CS). We omit scheduled CS as hospitals typically plan these births ahead and perform them well ahead of full term status. Thus, those births are less responsive to crowding and are performed under different circumstances than (attempted) spontaneous births (with a special focus of medical staff on planning them and on devoting extra resources to them).¹⁵ These constraints result in a final analysis sample of 796,416 mothers and their (singleton) births.

2.2.2. Outcome variables

Ward-level Adjustments. Faced with crowding, maternity ward staff can adjust their behavior along various margins, some of which affect the interpretation of our estimates and some of which we can directly study in this paper. First, they may call in additional staff; second, they may try to affect the pool of patients arriving at the ward by referring some, maybe more complicated, cases to other hospitals; third, they may change their procedure use to less labor-intensive procedures. First, maternity wards typically have a number of staff on hold and can call in additional midwives or physicians in case of crowding: Danish midwives face ex ante large uncertainty about the exact timing of their work hours: only 54 percent of their total hours worked are planned well ahead of time and as many as 15 percent of work hours consist of time being available from home. Thus, as we would expect, hospitals can buffer some of the observed crowding (Danske Regioner, 2017). Unfortunately, we have no hospital- or individuallevel data on staff or staff on hold but even with access to this type of information, the nature of the response-both in terms of the number of staff members called in and in terms of their skill set-would be endogenous to the severity of patient cases. Given that we consider the reduced form estimates of crowding, we likely underestimate the impact of crowding (that may be buffered by additional staff called into duty as an endogenous response to crowding).

Second, a channel of adjustment that we can study is the mothers' probability of being transferred to another hospital in the case of crowding. Aggregate data does not suggest that this channel is quantitatively important: A recent policy report from the Danish regions reports that women in active labor are only very rarely referred to other hospitals (Danish Regional Administration 2017): in 2013, in the Capital Region, out of about 20,000 births, 177 women were sent to other wards; in the Region of Southern Denmark, the corresponding numbers were as low as seven out of roughly 11,000. To explore the question as to whether transfer of births to other wards is impacted by crowding, we study the probability of being admitted to another hospital for birth than the one assigned in the first trimester of the pregnancy. This measure captures women who happened to be away from home at the start of labor, potential referrals to other wards because of crowding, and referrals during pregnancy, due to the need for specialized treatment. Therefore, this measure is also driven by other factors than ward responses to crowding.

Third, we study different types of procedure use. We exploit unique data on medical procedures and the timing of events during and after labor. We study mothers' probability of experiencing a (medical) stimulation during labor, their probability of experiencing an induction of labor, and the probability of experiencing an emergency CS.¹⁶ All these procedures require more continuous support and monitoring of the birth by staff (midwives and, in the case of a CS, also physicians). Changes in those procedures may also impact other outcomes, for instance inductions may lead to a higher risk of experiencing an emergency CS.

In a subset of data for the period 2011–2014, we can study a final set of procedural outcomes which are informative of staff- and resource-demanding factors: the timing between admission to hospital and the birth, the waiting time for an epidural, and the timely establishment of skin-to-skin contact. Later admissions during labor may be quantitatively more important than transfers of women during labor. If hospitals attempt to cope with crowding by delaying new admissions, we may expect shorter birth spells on crowded days. The establishment of skin-to-skin contact between infant and parents within 2 h after birth was a central quality indicator for Danish hospitals in the period that we study, and we therefore create an indicator for the ward meeting this target in the given birth.

Taken together, our data on procedures and waiting times allow us to examine a range of potential margins for adjustments made by the maternity ward staff in the case of crowding. Our discussion of these potential adjustments also illustrates why we examine the impact of crowding on health in a reduced form framework: Crowding likely affects many margins of staff behavior and goes beyond adjusting the staff patient ratio.

¹³ We have also generated crowding measures that exclude scheduled CS. Results are very similar and available on request.

¹⁴ We have also experimented with other treatment measures, e.g. the number of admissions on any admission day relative to the hospital- and year-specific median day and crowding variables relative to ward×year cells with very similar results.

¹⁵ In our main analyses, we control for the number of scheduled CS on the day of hospital admission.

¹⁶ In our data emergency CS are CS performed subsequent to an attempted vaginal birth or performed with short notice, i.e. the decision for a CS is taken less than 8 h prior to the procedure.

Health Outcomes. Having studied maternity ward procedure use, we turn to an analysis of measures of health at birth and in the longer run. As a first set of measures, we use data from the Danish Inpatient Register on diagnoses (at birth and in the post-birth period). Using diagnoses given around birth in our study is not without complications: It is not straightforward to classify birth complications by severity and their sensitivity to staff action (which would be the relevant complications to study). Finally, using the many available and fine-grained ICD diagnosis codes used to classify complications in our data may lead to Type I errors: testing hypotheses for multiple and small groups of diagnosis subgroups may lead us to make inappropriate conclusions about the power of our results.¹⁷

As a consequence, to use the administrative data on diagnosis and operations around birth while balancing the above factors, we proceed in two steps: First, we create broad summary measures for both maternal complications at birth and post-birth, i.e. indicators that are one for a broad group of diagnosis groups related to complications at birth and to post-birth complications.¹⁸ Second, we create three additional measures (two additional measures linked to birth and one additional post-birth measure) that allow us to zoom in on both the staff-intensity and severity of complications:¹⁹ For complications at birth, we single out one diagnosis group (and do not include it in our initial summary measure) for further scrutiny: Lacerations of varying degree. This measure is suitable for our analysis for three reasons: (i) it is closely monitored by the national Danish health agencies as a measure of hospital performance (thus hospitals and personnel put emphasis on monitoring it), (ii) it is modifiable by staff (midwife) attention, and (iii) it allows us to measure the severity of the complication (by focussing on either relatively mild degree one to two, or severe degree three to four lacerations). For post-birth complications, we create a measure of severity by creating a complication indicator that is one only for mothers, who both have a relevant diagnosis of post-birth complications and a relevant operation code in the 90 days post birth. The reasoning for this measure is that these mothers suffer more severe complications that require additional attention by medical professionals.

To complete our measures of health at birth, we report results for the child's probability of having a low APGAR score at 5 min. We both consider a cut-off of APGAR > 7 and APGAR > 9. For the child, we consider a number of additional health measures: First, we study mortality as the probability of a stillbirth, a perinatal death (stillbirth or death in first week) and first-year death. Second, we use measures of health care usage in the short- and longer run: We study the probability of neonatal care unit admission,²⁰ the length of hospital stay after birth, readmissions to hospital, and contacts with GPs within the first month and during the first and second year of the child's life.²¹

2.3. Descriptive statistics

Appendix Table A.1 provides summary statistics for the key variables in our analysis dataset. Excluding the focal mother, the average daily number of maternity ward admissions across all hospitals and years is 8.4 in our sample. The same statistic excluding scheduled CS is 7.6. The table shows large variation in the number of daily admissions caused by both differences across wards and days. Note that, by construction, busy days have more births (and thus more observations) and therefore the average percentile rank in the sample of births is above 0.5.

The lower part of Table A.1 describes our central outcome measures. On average, 19 percent of mothers experience an induction. Illustrating that we use a relatively broad measure of complications at birth, 57 percent of mothers experience a complication (see Appendix B for a list of the ICD codes included in this measure.) Five percent of mothers fall into the group with post-birth complications. Measures of child health show that readmissions to hospital in the first year are not rare (22 percent) while mortality is at very low levels.

When moving to more detailed procedural data, our data only covers the years 2011–2014 and thus the number of observations is smaller. 24 percent of mothers in our sample have an epidural for pain relief and their average waiting time for the epidural to be administered is around 37 min.

To describe our treatment variable of interest, crowding at the maternity ward, in the sample, Appendix Fig. A.1 panel (a) illustrates the distribution in the raw number of admissions for all wards for the year 2010. Maternity wards experience between one and 25 admissions per day (see also Appendix Table A.1). The median number of admissions in our sample is seven. In panel (b) we show a residualized measure of admissions for the year 2010 (our residualized measure takes out year, season, weekday, ward and ward \times year fixed effects, see the next section). The figure illustrates that although we account for those factors – e.g. that there are hospital wards of different size and that certain days of the week are more busy than others – there is remaining variation, namely days with deviations from the average day between –8 and 8 admissions.

Fig. 1 illustrates the number of admissions at the ward level. The top panel of Fig. 1 shows the daily admissions for the year 2010 for a small ward (Horsens) and Denmark's largest maternity ward (Hvidovre). These maternity wards

¹⁷ Another issue is registration practice: we may be concerned that either the focus on detailed registrations of diagnoses or the member of staff responsible for registration could be a function of crowding and thus be impacted. However, typically registrations for the Inpatient Registry are made after the birth is completed (while during the birth midwives make notes) and according to monitoring requirements imposed by the national authorities, hospitals are required to complete registrations for every birth event in the Inpatient Registry. We therefore assume that this issue should be minor.

¹⁸ For a list of the relevant ICD 10 groups, consult Appendix B.

¹⁹ We fully acknowledge that this choice of specific outcomes focuses our analyses to specific diagnosis groups and is not exhaustive.

²⁰ Our measure of neonatal care admission comes from the medical birth registry and is defined as admission to a specialized paediatric ward and a diagnosis code from IDC10 group "P".

²¹ For each outcome we omit children who die prior to the "measurement period", e.g. observations involving stillbirths are not included in the analysis sample for hospital admissions.



Fig. 1. Daily number of admissions and day-to-day variation in the number of daily admissions (residualized), selected maternity wards.

had on average, about five and 15 daily admissions, respectively. However, there is substantial variation in admissions in both the small and the large maternity ward. This point is underlined by the bottom panel of Fig. 1, which shows the relative day-to-day fluctuations in admissions for the same two wards. We observe that days with more than twice as many admissions than the day before are common in both the small and large maternity wards.²² Furthermore, we observe day-to-day changes in both the raw measure of number of admissions and in our residualized measure in the two wards.

3. Empirical methods

To examine the impact of temporary crowding on maternal and child health, we exploit residual variation in the number of admissions to Danish maternity wards. Specifically, we estimate the reduced form relationship

$$Y_{idsyw} = \alpha_0 + \alpha_1 crow d_{idsyw} + \boldsymbol{\beta}' X_{idsyw} + \boldsymbol{\delta}_{\boldsymbol{y}} + \boldsymbol{\lambda}_{\boldsymbol{s}} + \boldsymbol{\theta}_{\boldsymbol{d}} + \boldsymbol{\gamma}_{\boldsymbol{w}} + \boldsymbol{\delta}_{\boldsymbol{y}} \times \boldsymbol{\gamma}_{\boldsymbol{w}} + \boldsymbol{\epsilon}_{idsyw}$$
(1)

where Y is an outcome of interest for mother *i* admitted to ward *w* on weekday *d* in season (quarter) *s* of year *y*, such as the probability of the mother experiencing complications at birth. *crowd_{idsyw}* is the level of crowding at the ward on the day of admission measured as (1) the (leaveout) number of admissions on the day of admission or (2) the percentile rank of the admission day (within a given maternity ward, season and year) in terms of the (leaveout) number of admitted mothers.²³

The vector X_{idsyw} includes observable mother and father and an indicator for pregnancy complications.²⁴ Further-

²² This point is important because our sample of maternity wards consists of larger units in the end of the sample period compared to the initial years due to hospital mergers and an administrative reform in 2007.

²³ We have also computed alternative measures of crowding such as the (leave-out) number of admitted births relative to the median number of admitted women for a given maternity ward in a given year, or a moving average specification (i.e. computing crowding based on the previous day's and the current day's cumulated admissions). We discuss our results for alternative measures in Section 4.

²⁴ The exclusion of individual level control variables does not impact our conclusions and results for all main analyses excluding them are available on request.

more, we control for the number of scheduled CS on a given day and in a given ward.²⁵ To account for systematic differences across the population of mothers across years and season, we include vear of birth (δ_{v}) and season fixed effects λ_s . These cohort effects also capture the impact of shocks such as nation-wide changes in recommendations on procedure use. Furthermore, to capture systematic variation in birth-related admissions over weekdays and weekends, we include weekday indicators (θ_d). γ_w accounts for time-invariant differences across maternity wards. Finally, by including ward \times year fixed effects ($\delta_v \times \gamma_w$) we flexibly account for shocks specific to certain wards, such as changes in the population of mothers due to changes in catchment areas of hospitals or organizational changes within the ward.²⁶ The parameter of interest is α_1 , which measures the reduced form effect of crowding. We cluster standard errors at the ward level to capture arbitrary correlations in unobservable characteristics across mothers within the same hospital.

To ease comparison across the two treatment measures and to give a sense of the variation in the level of crowding that mothers potentially can experience around their own admission for birth, we calculate for each admission day a "potential range of crowding". This measure captures the variation around the actual admission day, which mothers cannot perfectly control. For each admission day we calculate the absolute difference in admissions between the most and least crowded day considering the admission day, the day before and the day after the admission, see Appendix Table A.2. In our sample, this difference, or the potential range of crowding, is around 3.4 on average, i.e. it suggests that on average women can expect the number of admissions to vary with three births, when we consider the days just around the mothers' actual admission day (this variation is slightly smaller when we consider our residualized measure taking out year, season, weekday, ward and ward \times year fixed effects, see Appendix Table A.2). Thus, when we discuss our main results for the absolute measure of crowding, we calculate the impact of an increase in admissions with three births.²⁷ Similarly, we compute the potential range of crowding based on our relative treatment measure. For the days around mothers' actual admission day, the average expectable variation is around 30 percentile ranks.

Identifying Assumptions To uncover a consistent estimate of α_1 , we assume that the residual variation in the number of admissions for childbirth at a given hospital is uncorrelated with unobservable characteristics that also impact outcomes. Importantly, this assumption postulates the absence of policy changes or other time-varying characteristics of women or hospitals that systematically covary with our variation in (temporary) crowding. We argue that it is reasonable that our set of fixed effects and controls for parental and pregnancy characteristics take care of this threat.²⁸

While this assumption is inherently untestable, Table 1 presents an informal assessment of the credibility of our design. Each row of Table 1 presents a series of estimation results for regressions of maternal, paternal and birth characteristics on our treatment variable, the (leave-out) number of admissions on the day of the admission of the mother. Moving from column (1) to (6) we add different sets of fixed effects to this regression. For our design to identify the effect of crowding, we expect predetermined characteristics of parents and the birth to be uncorrelated with our treatment variable once we account for factors that likely bias our estimation, such as a comparison between large (specialized) wards and small wards or different geographical locations.

As Table 1 shows, the regressions in column (1) show small albeit significant correlations for central observable characteristics and our treatment variable. These correlations decrease in size and loose significance when we move towards column (6) with the full set of fixed effects, our preferred specification.

Especially accounting for hospital and year fixed effects takes away predictive power of our treatment variable measure in our regressions with predetermined observable characteristics as outcomes. While our main estimate on mothers' income remains significant, the estimate is very small: One additional admission is associated with a 144 DKK (around 22 USD) difference in mothers' annual income. The significant effect of crowding on the probability of the father holding a university degree amounts to a 18 percent increase at the relevant sample mean.²⁹ Given few differences and our inclusion of parental characteristics as controls, our findings lend credibility to our identifying assumption that, conditional on the set of fixed effects and controlling for observable characteristics, we identify the impact of crowding on outcomes.

²⁵ Recall that those CS are included to calculate our measures of crowding but are omitted in the analysis.

²⁶ Such changes occur for at least two reasons: First, in 2007 the former 16 Danish counties were merged into five regions. As hospitals were administered by the counties until 2007 and since then by the new regions, both the organization and the catchment areas were potentially changed by this reform. Second, closures of maternity wards during the period that we consider affect nearby wards through the number and composition of patients. Such changes are absorbed by the year-specific maternity ward fixed effects and we further assess the robustness of our results by restricting the sample to wards which do not experience large changes in admission numbers in Section 4.

²⁷ Appendix Table A.2 also reports the 75th and 90th percentile of the distribution of the potential range of crowding to illustrate the effect size for mothers being admitted on a day with an unusually high level of crowding compared to close by days.

²⁸ Additional evidence to substantiate this claim comes from analyses that exclude wards that merge during our time period (a time-varying characteristic that may be of importance). As we detail in Section 4, our analysis and conclusions are not impacted by this sample constraint.

²⁹ All conclusions also hold for our relative measures of crowding, as reported in Appendix Table A.3. Importantly, the nature of potential biases is different when applying the different crowding measures, especially in column 1 where for the absolute measure, the regression compares primarily across small and large wards, while the percentile measure abstracts from this comparison being based on a calculation inside a given ward and year.

The effect of crowding (absolute measure) on pre-determined characteristics, 2000-2014.

	(1)	(2)	(3)	(4)	(5)	(6)	Ν
Mother income	1.357***	0.517*	0.312***	0.312*	0.336***	0.144***	773,173
	(0.392)	(0.190)	(0.114)	(0.115)	(0.122)	(0.046)	
Mother university deg.	0.951***	0.221*	0.084	0.085	0.109	0.011	796,416
	(0.185)	(0.106)	(0.057)	(0.057)	(0.065)	(0.012)	
Mother age	0.064***	0.027***	0.017***	0.018***	0.017***	0.004*	773,173
	(0.017)	(0.007)	(0.003)	(0.003)	(0.004)	(0.002)	
Father income	1.151*	0.393*	0.376*	0.371*	0.406*	-0.011	765,675
	(0.582)	(0.201)	(0.171)	(0.174)	(0.180)	(0.076)	
Father university deg.	0.958***	0.182*	0.093*	0.092*	0.113*	0.026***	786,673
	(0.187)	(0.079)	(0.045)	(0.045)	(0.050)	(0.009)	
Father age	0.049***	0.021***	0.012***	0.013***	0.013***	0.002	765,675
	(0.016)	(0.007)	(0.003)	(0.003)	(0.003)	(0.002)	
Mother GP contacts	0.004	0.046***	0.002	0.003	-0.007	-0.003	775,943
	(0.023)	(0.015)	(0.005)	(0.005)	(0.005)	(0.003)	
Mother GP fees	-1.964***	-2.286*	0.057	0.067	0.011	-0.024	775,943
	(0.516)	(0.900)	(0.044)	(0.044)	(0.047)	(0.025)	
Birth weight (grams)	-3.081***	-0.200	0.285	0.154	0.303	-0.219	786,530
	(0.608)	(0.363)	(0.271)	(0.269)	(0.293)	(0.348)	
Hospital FE		Yes	Yes	Yes	Yes	Yes	
Year FE			Yes	Yes	Yes	Yes	
Quarter of Year FE				Yes	Yes	Yes	
Day of week FE					Yes	Yes	
Hospital × year FE						Yes	

Notes: Each cell presents point estimates from a separate regression. For a description of the main sample, see Section 2.2. All parental characteristics are measured two years prior to the birth of the focal child. Maternal GP contacts and the fees related to them are measured during a one year period. The outcome variable for all regressions in a given row is denoted in the first column. Standard errors are clustered at the hospital level and presented in parentheses. The table shows estimates for the absolute crowding measure (absolute number of admissions at a given ward and admission date). For an equivalent table based on the percentile rank, see Appendix Table A.3. The changing number of observations across rows (outcomes) is related to missing parental characteristics by including indicators for missing data in our main specifications. Significance levels: * p < 0.1; ** p < 0.05; *** p < 0.01.

4. Results

In this section, we present our main results for the impact of crowding on ward-level adjustments. We then proceed to studying consequences of admission on a crowded day for maternal and infant health outcomes. In all tables, each cell shows point estimates and standard errors from a different regression. Columns 1 and 2 present the results for our two different crowding measures, respectively: the total number of admissions in the same ward and day and the percentile rank of the admission day (within hospital, season and year cells). All regressions are based on the full set of fixed effects and controls, as presented in equation (1). The bottom row of each table shows the mean range in the level of crowding, i.e. the range of crowding typically observed around a given admission day, see also Appendix Table A.2. In all tables, coefficients and control means for indicator variables are pre-multiplied with 100 to ease readability.

4.1. Maternity ward adjustments to crowding

Our analyses for procedure use and timing focus on potential adjustments and changes inside the maternity ward. A first potential adjustment of maternity wards is to direct patients to other, less crowded wards. As described in the data section, we do not observe mothers who call their ward while already in labor and who may then be directed to a different, less crowded ward. While other sources suggest that the transfer of women during labor between wards is quantitatively unimportant, we still attempt to study this margin. Thus (in the absence of data on phone contact with the designated maternity ward), we study the probability of giving birth at a different ward than the one assigned during the first trimester. As Table 2 illustrates, 18 percent of the women in our analysis sample do not give birth at their initially assigned ward. When we examine the impact of crowding on this measure, we do not find that crowding impacts this probability of being admitted to a different ward. While we think that this analysis suggests that transfers during labor are not driving our results, a caveat is that the majority of women who give birth at a different hospital than the initial assigned hospital are likely to be "scheduled movers", i.e. women who are transferred to specialized maternity wards prior to birth.

Next, we consider other channels for maternity ward adjustments. Specifically, we focus on typical procedures used during labor and observed in the medical records. We find that on crowded days maternity wards are less likely to perform inductions and stimulations, albeit the effects are small. Three additional birth admissions (the range of crowding) are related to a $3 \times 0.225 = 0.7$ percentage points difference in the probability of an induction. At the sample mean induction rate of 18.6 percent, this corresponds to about a four percent decrease.³⁰ Similarly, for the percentile rank measure, a change of 30 percentile ranks in crowding (which is the average variation around admission

 $^{^{30}}$ For mothers entering at days with unusually high levels of crowding the effect is a 7 × 0.225 = 1.6 percentage points difference in the probability of an induction, see Appendix Table A.2.

The effect of crowding on transfers and procedure use, 2000–2014.

	Absolute (1)	Percentile (2)	Mean of dep. var	Ν
Admitted to other hosp than first hosp	0.006 (0.009)	-0.168 (0.130)	18.212	785,336
Stimulation of labor	-0.120*** (0.018)	-1.170*** (0.238)	28.055	796,416
Induction	-0.225*** (0.023)	-2.326*** (0.191)	18.612	796,416
Emergency CS	0.002 (0.014)	0.087 (0.136)	12.022	796,416
Mean of 'expected' variation in crowding	3.047	0.298		

Notes: Each cell presents point estimates from a separate regression. The first column presents the estimate for the impact of the absolute number of admissions, column (2) presents the estimate for the impact of the percentile rank of the admission day in the distribution of days in the hospital and year. All coefficients come from regressions accounting for fixed effects for the maternity ward (hospital), year, season, day of week, and ward × year, as well as the following set of control variables: the number of scheduled CS at the hospital and day of admission, an indicator for pregnancy complications, maternal and paternal and paternal and paternal education (higher education, university degree), indicators for maternal and paternal and paternal disability pension status, maternal and paternal education vs. not in education), indicators for maternal and paternal region non-western origin, and separate indicators that are one for individuals with missing values for the parental control variables. Standard errors are clustered at the hospital level. The mean of the variation in crowding is the mean range in crowding across the mother's day of admission, the day after (for more details, see Appendix Table A.2).

Significance levels: * *p* < 0.1; ** *p* < 0.05; *** *p* < 0.01.

Table 3

The effect of crowding on procedures at birth; constrained sample with timing and procedure data, 2011–2014.

Absolute (1)	Percentile (2)	Mean of dep. var	Ν
-0.208***	-2.150***	50.897	182,564
(0.036)	(0.361)		
-0.136***	-1.514***	24.178	192,748
(0.036)	(0.377)		
-0.017	-0.190	76.462	192,748
(0.041)	(0.388)		
0.056	-0.133	48.274	41,430
(0.087)	(0.917)		
3.263	0.290		
	Absolute (1) -0.208*** (0.036) -0.136*** (0.036) -0.017 (0.041) 0.056 (0.087) 3.263	Absolute (1) Percentile (2) -0.208*** -2.150*** (0.036) (0.361) -0.136*** -1.514*** (0.036) (0.377) -0.017 -0.190 (0.041) (0.388) 0.056 -0.133 (0.087) (0.917) 3.263 0.290	Absolute (1) Percentile (2) Mean of dep. var -0.208*** -2.150*** 50.897 (0.036) (0.361) -0.136*** -1.514*** 24.178 (0.036) (0.377) -0.017 -0.190 76.462 (0.041) (0.388) -0.056 -0.133 48.274 (0.087) (0.917) -3.263 0.290

Notes: See notes for Table 2.

Significance levels: * *p* < 0.1; ** *p* < 0.05; *** *p* < 0.01.

days in our sample) corresponds to a $0.298 \times 2.3 = 0.7$ percentage points difference in the probability of an induction (or a difference of about four percent, evaluated at the sample mean). The effects are thus very similar regardless of whether we define the treatment as the number of admissions or relative to the number of admissions inside the ward, season and year.

Finally, Table 3 constrains our analysis sample to the most recent years (2011-2014) and examines outcomes indicative of the access to resources in the maternity ward for women. We analyze the impact of crowding on the time between admission time and birth, the propensity to receive an epidural, the waiting time for an epidural conditional on having one, and an indicator for skin-toskin contact. While estimates are again small, we find that women admitted on more crowded days are less likely to have above-median duration ward stays, i.e. most likely arrive later at the ward and give birth faster. Given that women are typically in contact with their ward in the hours up to hospital admission, we take this finding as indication for wards delaying admissions during peak hours. After admission, we find that women admitted on more crowded days are less likely to have an epidural for pain relief. One reason for this result may be that maternity wards delay the admission of women. Thus women, who are admitted closer to their actual birth, will not be able to get an epidural (given the waiting time related to the admission of epidurals by physicians). At the same time, in line with wards being able to buffer temporary crowding, we do not find longer waiting times for actual epidurals and no clear indication for crowding impacting a measure of birth experience quality, namely the timely establishment of skin-to-skin contact between parents and child.

4.2. Mother and child health

Table 4 presents our main results for the effects of temporary crowding on the day of admission for birth on health outcomes.³¹ In panel (A) we first report our results for birth

³¹ To show that our findings are not driven by the choice of data period, Appendix Tables A.4 and A.5 replicate all our main results for the shorter sample of data from the years 2011–2014 used in our analyses on specific procedures. We find very similar results (albeit averages for the dependent variables increase illustrating the importance of control for year fixed effects): Effects for inductions and stimulations are slightly larger and we do not find a significant effect on the aggregate measure of complications

The effect of crowding on birth outcomes and child health, 2000–2014.

	Absolute (1)	Percentile (2)	Mean of dep. var	Ν
A. Birth outcomes				
Comp. at birth	-0.049^{***}	-0.466^{***}	57.084	796,416
*	(0.012)	(0.162)		
Comp. at birth: Laceration degree 1 or 2	-0.074***	-0.789***	34.899	796,416
	(0.017)	(0.194)		
Comp. at birth: Laceration degree 3 or 4	0.007	0.114	3.164	796,416
	(0.008)	(0.078)		,
APGAR>7	-0.007	-0.046	97.642	796,416
	(0.008)	(0.080)		
APGAR>9	0.023	0.091	91.467	796,416
	(0.018)	(0.168)		
Post-birth complications, mom	-0.001	-0.019	4.889	796,416
r · · · · · ·	(0.006)	(0.083)		,
Post-birth complications – severe, mom	-0.014***	-0.160**	2.632	796.416
r · · · · · · · · · · · · · · · · · · ·	(0.004)	(0.059)		, .
B. Child health outcomes				
Hospital nights at birth	0.001	-0.003	3.470	791,290
1 0	(0.004)	(0.039)		
Neonatal care admission	0.007	0.087	8.052	796,416
	(0.015)	(0.150)		,
Readmitted first 28 days	-0.015*	-0.140	5.409	792.754
, and the second s	(0.008)	(0.091)		- , -
Readmitted first year	-0.012	-0.155	22.439	791,772
5	(0.017)	(0.162)		
Readmitted second year	-0.018	-0.227	16.249	791,772
5	(0.018)	(0.211)		
Contacts with gp first month	0.001	0.005	0.700	792,754
01	(0.001)	(0.006)		,
Contacts with gp first year	-0.010	-0.104	13.301	791,772
	(0.006)	(0.071)		
Contacts with gp second year	-0.008**	-0.068	7.756	791,772
<u> </u>	(0.003)	(0.049)		-
Mean of 'expected' variation in crowding	3.047	0.298		

Notes: See notes for Table 2.

Significance levels: * *p* < 0.1; ** *p* < 0.05; *** *p* < 0.01.

outcomes and in panel (B) we extend the analysis to consider outcomes related to child health (after birth).

Relating the point estimates in panel (A) of Table 4 to the mean range in level of crowding, we observe that an additional three admitted births result in a very small ($3 \times 0.05 = 0.15$ percentage point) difference in the probability of complications at birth.³² Zooming in on complications that may be the result of staff shortage, we study the probability of mild and serious lacerations. While we find a significant and negative impact of crowding on mild lacerations, we do not find an effect of crowding on severe lacerations.³³ Given that lacerations of very mild form may be subject to staff assessment, we cannot rule out that crowding impacts their registrations.

Considering other relevant birth outcomes, our estimates for both the probability of the child's APGAR score being low and the probability of the mother experiencing post-birth complications are very small and do not sug-

gest an impact of temporary crowding on these outcomes. Focusing on severe maternal post-birth complications, we even find a small negative effect of crowding (an estimate that implies roughly a 1.5 percent decrease in the probability of severe complications for our absolute measure of crowding). These last findings are different from results in Avdic et al. (2018), who find that hospital mergers increase mothers' probability of experiencing post-birth trauma, a finding that they attribute to potential crowding at merged wards. These differences may be due to (at least) two factors. First, our inside-ward variation in crowding may be less drastic and disruptive than a ward merger. Second, we do not find an impact of crowding on the probability of experiencing an unscheduled (emergency) CS. We find decreased procedure use on crowded days, which may reduce the need for follow up treatments due to side effects of procedures such as pain relief. This difference may be particularly important for maternal postnatal health.

In panel (B) of Table 4, we extend our analyses to postbirth child health outcomes measured as health care usage at either the GP or at the hospital (admissions).³⁴ Across

at birth. However, we confirm the main patterns in our findings for post birth complications (for severe lacerations).

 $^{^{32}}$ Similarly, a 30 percentile rank variation in crowding results in a 0.29×0.5 = 0.15 percentage points difference in the probability of birth complications.

³³ Recall that lacerations are excluded from our overall complications measure.

³⁴ We have also considered the number of nights spent at the neonatal care unit. We find no significant impacts of crowding at this margin and results are available on request.



Fig. 2. The effect of the number of daily admissions on procedure use and health outcomes, local linear regressions. *Notes*: The plots show the relationships between the residualized *y*-variable and the residualized number of admissions to maternity wards. Each marker contains two percent of observations and plots the mean of the *y*-variable against the mean of the *x*-variable. The line represents the local linear regression, with a bandwidth of one.

specifications, we find very small and mostly precise estimates that suggest no important longer-run impact of temporary crowding on the health care usage of children in our sample. As in panel (A) the effect sizes are very similar across treatment measures once the variability in crowding is taken into account.

Finally, we have considered the impact of crowding on the probability of the birth being a stillbirth, resulting in a perinatal death (including the first week of life) and the probability of a first-year death.³⁵ As shown in Appendix Table A.6, we do not find any large impact of crowding on these rare and severe health outcomes. Estimates are very small and thus do not support an impact of (temporary) crowding at Danish maternity wards for our general population of (non-scheduled) births.

In additional robustness analyses we confirm that our results are not driven by ward mergers (Appendix Table A.7 with estimates based on stable wards only and excluding wards that merge throughout the sample period). Further our findings are unchanged when we measure crowding based on the day of admission as well as the day prior to admission (Appendix Table A.8).³⁶ While point estimates

³⁵ Results for perinatal deaths including the first two weeks of life are very similar and available on request.

³⁶ Appendix Table A.7 presents results for two crowding measures only based on absolute number of admissions (to ease comparison to the main specification): A two day average measure of crowding and changes in crowding between the day of admission and the day prior to admission.

Heterogeneity: The effect of crowding on selected health outcomes and use of procedures, 2000-2014.

	Main	Gest. age		Parity		Hospital size	
	(1)	<41w (2)	≥41w (3)	=1 (4)	>1 (5)	<p50 (6)</p50 	>p50 (7)
Stimulation of labor	-0.120*** (0.018)	-0.113*** (0.025) [0.824]	-0.103** (0.043)	-0.172*** (0.056) [0.231]	-0.110*** (0.021)	-0.087** (0.034) [0.295]	-0.124*** (0.014)
Induction	-0.225*** (0.023)	-0.180*** (0.019) [0.003]	-0.281*** (0.044)	-0.287*** (0.063) [0.059]	-0.215*** (0.025)	-0.176*** (0.022) [0.001]	-0.268*** (0.027)
Complications at birth	-0.049*** (0.012)	-0.060*** (0.017) [0.096]	0.005 (0.036)	-0.090 (0.063) [0.371]	-0.042*** (0.010)	-0.032** (0.014) [0.731]	-0.044** (0.020)
Complications at birth: Laceration degree 3 or 4	0.007 (0.008)	0.015 (0.010) [0.130]	-0.012 (0.017)	0.013 (0.014) [0.733]	0.007 (0.009)	0.001 (0.018) [0.310]	0.016** (0.007)
APGAR>7	-0.007 (0.008)	0.001 (0.009)	-0.010 (0.011)	-0.042* (0.023)	-0.001 (0.008) 241	0.007 (0.013)	-0.019** (0.008)
Post-birth complications, mom	-0.001 (0.006)	0.010 (0.009) [0.061]	-0.028* (0.016)	0.005 (0.016) [0.730]	-0.002 (0.006)	-0.001 (0.009) [0.816]	0.003 (0.008)
Post-birth complications – severe, mom	-0.014^{***} (0.004)	-0.006 (0.007) [0.077]	-0.033*** (0.012)	-0.026* (0.014) [0.372]	-0.012^{***} (0.003)	-0.010 (0.008) [0.769]	-0.014** (0.006)
Observations	796,416	577,158	217,031	200,871	595,545	361,702	434,714

Notes: Each cell presents point estimates from a separate regression for the impact of the number of admissions. Column (1) is based on the full sample, columns (2) and (3) are for children born below and above 41 completed weeks of gestation, columns (4) and (5) are for first-born and higher parity samples, columns (6) and (7) are for small and large hospitals, respectively (split at the median). P-values for the test of equal coefficients across subgroups are in square brackets. Standard errors are clustered at the hospital level and presented in parenthesis. For further details see notes for Table 2. Significance levels: * p < 0.1; ** p < 0.05; *** p < 0.01.

and significance change marginally, our main conclusions hold across the different definitions of sample and crowding measure.

In sum, our estimates for the effects of temporary crowding on health at birth have the same sign across different measures of crowding. While our large sample results in precise estimates, the size of the estimates does not indicate large health effects and suggests that wards are able to accommodate to the observed variation in daily admissions without health risks for a general population of mothers and their infants.

As our findings suggest no or very few health effects on more crowded days, our results for inductions may indicate that a decreased use (or delay) of marginal inductions is not hurtful in this population of births (a small decrease in maternal severe post-birth complications may suggest the opposite even). Inductions have been shown to be correlated with more complicated progression of labor and (in some studies) with the increased use of emergency CS. As we show below, we find that the probability of experiencing an induction decreases on crowded days for both pregnancies prior to and after completed week 41 (which is a relevant cut-off in most Danish hospitals for considering an induction for safety reasons). Thus, our results suggest that crowding may delay and potentially prevent non-medically indicated inductions.

4.3. Are effects asymmetric or heterogeneous?

To assess whether our linear model appropriately captures the underlying relationship between temporary crowding and our outcomes, Fig. 2 plots the relationship between our residualized treatment variable and residualized outcome measures. As the figures show, the linear specification appears to be appropriate for all considered outcomes. This finding suggests that there is no "threshold" for the effects of crowding, at least for the variation that we observe in our data.

Finally, we assess the heterogeneity in our reported results along three dimensions using the absolute level of crowding as the treatment variable.³⁷ As Table 5 illustrates, overall we see limited evidence for heterogeneous impacts of crowding along the dimensions that we consider. First, in columns (2) and (3) of Table 5 we split the sample by gestational age (below full 41 weeks or not). While most coefficients are similar across these two groups, the effect of crowding on the probability of inductions is significantly higher for births after week 41. In columns (4) and (5) we split the sample by parity. The effect of crowding on the probability of induction is largest for first-born children (marginally significant). Finally, in columns (6) and (7) we split the sample by maternity ward size. The effect of crowding on the probability of an induction is significantly larger in large wards. There is some indication for a

In both columns we interpret estimates as the effect of an additional admission.

³⁷ We only report the results for our main outcomes—additional results are available on request.

negative effect on the probability of having an APGAR score >7 in large wards. Again, this estimate is very small at the relevant mean.

5. Conclusion

While we have solid evidence for the benefit of medical care for at-risk births in the short and long run, we lack knowledge on the impact of variability of care for a more general population of births. This paper has studied the impact of exogenous shocks to maternity wards unexpected variation in the number of daily admissions—on the inside-ward change in procedure use for this group of patients and their health outcomes. Rather than exploiting variation across hospitals, we have isolated residual variation in daily admissions net of maternity ward, year, season, day of week and ward × year fixed effects. Using Danish administrative data, we have assessed several relevant margins of procedure use and health.

Importantly, our results do not indicate large negative impacts on health at birth and in the short-run due to variation in maternity ward crowding for a generally healthy population of mothers and their infants. This result echoes earlier findings for at-risk newborns and is in line with other causal estimates for the effects of crowding on patient health from other settings. However, as opposed to earlier work, we document that maternity wards adapt their allocation of procedures and care also for a general population of births (i.e. a relatively healthy patient group): Due to temporary crowding, they allocate fewer procedures to mothers during birth and admit them later during labor. At the same time, we document that crowding does not impact the handling of complications during labor when measured as the probability of having an emergency CS. These changes of allocation of care do not seem to impact child health but may even lead to fewer severe post-birth complications for mothers.

Our findings emphasize the benefits of collecting rigorous evidence on the impact of crowding across different settings and patient groups in the health care system. This evidence can inform policies about the importance of crowding for medical care decisions and patient health and likely leads to country- and setting-specific recommendations. Importantly, exploiting temporary variation in crowding, our study is not informative about the optimal level of maternity ward care. Furthermore, given our identifying variation and the scheduling of the vast majority of complicated births as caesarean sections, our results do not carry over to at-risk births. However, and important for the prioritization of resources at the maternity ward, our main findings show that Danish maternity wards can buffer the impact of crowding at the observed levels without important health consequences for a general population of mothers and newborns.

Authors' contributions

Jonas Maibom, Hans Henrik Sievertsen, Marianne Simonsen and Miriam Wüst: Conceptualization, Methodology, Data curation, Writing, Reviewing and Editing.

Appendix A. Additional figures and tables



Fig. A.1. Distribution of the variation in the number of daily admissions to all Danish hospital wards, based on admissions in the year 2010.

J. Maibom, H.H. Sievertsen, M. Simonsen et al.

Table A.1

Summary statistics, means, standard deviations and distribution measures.

N. adm. same day 8.36 5.29 796,416 2.00 7.00 16.00 N. adm same day, no planned CS 7.57 4.71 796,416 2.00 7.00 14.00 Crowding percentile 0.60 0.27 796,416 0.20 0.64 0.95 Relative relative to median 1.26 0.62 796,416 0.60 0.17 2.00 Female 0.49 0.50 796,416 0.60 0.00 1.00 Gestational age 278.94 13.26 794,189 266.00 281.00 292.00 Birth after week 41 (over term) 0.27 0.45 794,189 0.00 0.00 1.00 Birth weight 3525.75 561.48 789,713 2870 3544 4200 Mother income (thousands) 178.63 100.31 773,173 64.94 172.27 288.84 Mother with university degree 0.13 0.34 796,416 0.00 0.00 1.00 Mother with university degree 0.23 <td< th=""><th></th><th>Mean</th><th>SD</th><th>Ν</th><th>P10</th><th>Median</th><th>P90</th></td<>		Mean	SD	Ν	P10	Median	P90
N. adm same day, no planned CS7.574.71796,4162.007.0014.00Crowding percentile0.600.27796,4160.200.640.95Relative relative to median1.260.62796,4160.601.172.00Female0.490.50796,4160.000.001.00Gestational age278.9413.26794,1890.000.001.00Birth after week 41 (over term)0.270.45794,1890.000.001.00First-born child0.460.50783,7620.000.001.00Birth weight3525.75561.48789,713287035444200Mother uncome (thousands)178.63100.31773,17364.94172.27288.84Mother with university degree0.130.34796,4160.000.001.00Mother with higher education0.230.42796,4160.000.001.00Mother enrolled in edu0.050.22796,4160.000.000.00Mother enrolled in edu0.050.22796,4160.000.000.00Father with university degree0.140.34765,67588.67234.19402.44Father with university degree0.140.34786,6730.000.001.00Father with higher education0.170.38786,6730.000.001.00Father with higher education0.170.38<	N. adm. same day	8.36	5.29	796,416	2.00	7.00	16.00
Crowding percentile0.600.27796,4160.200.640.95Relative relative to median1.260.62796,4160.601.172.00Female0.490.50796,4160.000.001.00Gestational age278.9413.26794,189266.00281.00292.00Birth after week 41 (over term)0.270.45794,1890.000.001.00First-born child0.460.50783,7620.000.001.00Birth weight3525.75561.48789,713287035444200Mother income (thousands)178.63100.31773,17364.94172.27288.84Mother western origin0.890.31773,1730.001.001.00Mother with university degree0.130.34796,4160.000.001.00Mother with higher education0.230.42796,4160.000.000.00Mother enrolled in edu0.050.22796,4160.000.000.00Mother enrolled in edu0.050.22796,4160.000.000.00Father western origin0.890.31765,67588.67234.19402.44Father western origin0.890.31765,67588.67234.19402.44Father western origin0.890.31765,67588.67234.19402.44Father with university degree0.140.34786,673<	N. adm same day, no planned CS	7.57	4.71	796,416	2.00	7.00	14.00
Relative relative to median1.260.62796,4160.601.172.00Female0.490.50796,4160.000.001.00Gestational age278.9413.26794,189266.00281.00292.00Birth after week 41 (over term)0.270.45794,1890.000.001.00First-born child0.460.50783,7620.000.001.00Birth weight3525.75561.48789,713287035444200Mother income (thousands)178.63100.31773,17364.94172.27288.84Mother western origin0.890.31773,1730.001.001.00Mother with university degree0.130.34796,4160.000.001.00Mother enrolled in edu0.050.22796,4160.000.000.00Mother enrolled in edu0.050.22796,4160.000.000.00Father western origin0.890.31765,67588.67234.19402.44Father western origin0.890.31765,67588.67234.19402.44Father western origin0.890.31765,67588.67234.19402.44Father western origin0.890.31765,67588.67234.19402.44Father western origin0.890.31765,67588.67234.19402.44Father with university degree0.140.34786	Crowding percentile	0.60	0.27	796,416	0.20	0.64	0.95
Female0.490.50796,4160.000.001.00Gestational age278.9413.26794,189266.00281.00292.00Birth after week 41 (over term)0.270.45794,1890.000.001.00First-born child0.460.50783,7620.000.001.00Birth weight3525.75561.48789,713287035444200Mother income (thousands)178.63100.31773,17364.94172.2728.84Mother with university degree0.130.34796,4160.000.001.00Mother with higher education0.230.42796,4160.000.001.00Mother enrolled in edu0.050.22796,4160.000.000.00Father income (thousands)248.85170.84765,67588.67234.19402.44Father western origin0.890.31765,67580.000.001.00Father western origin0.890.31765,67588.67234.19402.44Father western origin0.890.31765,67580.000.001.00Father western origin0.890.31765,6750.001.001.00Father western origin0.890.31765,6750.001.001.00Father western origin0.890.31765,6750.001.001.00Father with university degree0.140.34786,673 </td <td>Relative relative to median</td> <td>1.26</td> <td>0.62</td> <td>796,416</td> <td>0.60</td> <td>1.17</td> <td>2.00</td>	Relative relative to median	1.26	0.62	796,416	0.60	1.17	2.00
Gestational age278.9413.26794,189266.00281.00292.00Birth after week 41 (over term)0.270.45794,1890.000.001.00First-born child0.460.50783,7620.000.001.00Birth weight3525.75561.48789,713287035444200Mother income (thousands)178.63100.31773,17364.94172.2728.84Mother with university degree0.130.34796,4160.000.001.00Mother with higher education0.230.42796,4160.000.001.00Mother enrolled in edu0.050.22796,4160.000.000.00Father income (thousands)248.85170.84765,67588.67234.19402.44Father western origin0.890.31765,6750.001.001.00Mother enrolled in edu0.050.22796,4160.000.000.00Father income (thousands)248.85170.84765,67588.67234.19402.44Father western origin0.890.31765,6750.001.001.00Father with university degree0.140.34786,6730.000.001.00Father with higher education0.170.38786,6730.000.001.00Father with higher education0.010.08786,6730.000.001.00	Female	0.49	0.50	796,416	0.00	0.00	1.00
Birth after week 41 (over term)0.270.45794,1890.000.001.00First-born child0.460.50783,7620.000.001.00Birth weight3525.75561.48789,713287035444200Mother income (thousands)178.63100.31773,17364.94172.27288.84Mother western origin0.890.31773,1730.001.001.00Mother with university degree0.130.34796,4160.000.001.00Mother with higher education0.230.42796,4160.000.000.00Mother enrolled in edu0.050.22796,4160.000.000.00Father income (thousands)248.85170.84765,67588.67234.19402.44Father western origin0.890.31765,6750.001.001.00Father with university degree0.140.34786,6730.000.001.00Father with higher education0.170.38786,6730.000.001.00	Gestational age	278.94	13.26	794,189	266.00	281.00	292.00
First-born child0.460.50783,7620.000.001.00Birth weight3525.75561.48789,713287035444200Mother income (thousands)178.63100.31773,17364.94172.27288.84Mother western origin0.890.31773,1730.001.001.00Mother with university degree0.130.34796,4160.000.001.00Mother with higher education0.230.42796,4160.000.000.00Mother enrolled in edu0.050.22796,4160.000.000.00Father income (thousands)248.85170.84765,67588.67234.19402.44Father western origin0.890.31765,6750.001.001.00Father with university degree0.140.34786,6730.000.001.00Father with higher education0.170.38786,6730.000.001.00Father with higher education0.010.08786,6730.000.001.00	Birth after week 41 (over term)	0.27	0.45	794,189	0.00	0.00	1.00
Birth weight3525.75561.48789,713287035444200Mother income (thousands)178.63100.31773,17364.94172.27288.84Mother western origin0.890.31773,1730.001.001.00Mother with university degree0.130.34796,4160.000.001.00Mother with higher education0.230.42796,4160.000.000.00Mother enrolled in edu0.050.22796,4160.000.000.00Mother enrolled in edu0.050.22796,4160.000.000.00Father income (thousands)248.85170.84765,67588.67234.19402.44Father with university degree0.140.34786,6730.000.001.00Father with higher education0.170.38786,6730.000.001.00Father with higher education0.010.08786,6730.000.000.00	First-born child	0.46	0.50	783,762	0.00	0.00	1.00
Mother income (thousands)178.63100.31773,17364.94172.27288.84Mother western origin0.890.31773,1730.001.001.00Mother with university degree0.130.34796,4160.000.001.00Mother with higher education0.230.42796,4160.000.001.00Mother disability pension0.000.06796,4160.000.000.00Mother enrolled in edu0.050.22796,4160.000.000.00Father income (thousands)248.85170.84765,67588.67234.19402.44Father western origin0.890.31765,6750.001.001.00Father with university degree0.140.34786,6730.000.001.00Father with higher education0.170.38786,6730.000.001.00Father disability pension0.010.08786,6730.000.001.00	Birth weight	3525.75	561.48	789,713	2870	3544	4200
Mother western origin0.890.31773,1730.001.001.00Mother with university degree0.130.34796,4160.000.001.00Mother with higher education0.230.42796,4160.000.001.00Mother disability pension0.000.06796,4160.000.000.00Mother enrolled in edu0.050.22796,4160.000.000.00Father income (thousands)248.85170.84765,67588.67234.19402.44Father with university degree0.140.34786,6730.000.001.00Father with higher education0.170.38786,6730.000.001.00Father with begins on0.010.08786,6730.000.000.00	Mother income (thousands)	178.63	100.31	773,173	64.94	172.27	288.84
Mother with university degree0.130.34796,4160.000.001.00Mother with higher education0.230.42796,4160.000.001.00Mother disability pension0.000.06796,4160.000.000.00Mother enrolled in edu0.050.22796,4160.000.000.00Father income (thousands)248.85170.84765,67588.67234.19402.44Father western origin0.890.31765,6750.001.001.00Father with higher education0.170.38786,6730.000.001.00Father with bigher education0.010.08786,6730.000.000.00	Mother western origin	0.89	0.31	773,173	0.00	1.00	1.00
Mother with higher education0.230.42796,4160.000.001.00Mother disability pension0.000.06796,4160.000.000.00Mother enrolled in edu0.050.22796,4160.000.000.00Father income (thousands)248.85170.84765,67588.67234.19402.44Father western origin0.890.31765,6750.001.001.00Father with higher education0.170.38786,6730.000.001.00Father with disability pension0.010.08786,6730.000.000.00	Mother with university degree	0.13	0.34	796,416	0.00	0.00	1.00
Mother disability pension 0.00 0.06 796,416 0.00 0.00 0.00 Mother enrolled in edu 0.05 0.22 796,416 0.00 0.00 0.00 Father income (thousands) 248.85 170.84 765,675 88.67 234.19 402.44 Father western origin 0.89 0.31 765,675 0.00 1.00 1.00 Father with university degree 0.14 0.34 786,673 0.00 0.00 1.00 Father with higher education 0.17 0.38 786,673 0.00 0.00 1.00 Father disability pension 0.01 0.08 786,673 0.00 0.00 0.00	Mother with higher education	0.23	0.42	796,416	0.00	0.00	1.00
Mother enrolled in edu 0.05 0.22 796,416 0.00 0.00 0.00 Father income (thousands) 248.85 170.84 765,675 88.67 234.19 402.44 Father western origin 0.89 0.31 765,675 0.00 1.00 1.00 Father with university degree 0.14 0.34 786,673 0.00 0.00 1.00 Father with higher education 0.17 0.38 786,673 0.00 0.00 1.00 Father disability pension 0.01 0.08 786,673 0.00 0.00 0.00	Mother disability pension	0.00	0.06	796,416	0.00	0.00	0.00
Father income (thousands)248.85170.84765,67588.67234.19402.44Father western origin0.890.31765,6750.001.001.00Father with university degree0.140.34786,6730.000.001.00Father with higher education0.170.38786,6730.000.001.00Father disability pension0.010.08786,6730.000.000.00	Mother enrolled in edu	0.05	0.22	796,416	0.00	0.00	0.00
Father western origin 0.89 0.31 765,675 0.00 1.00 1.00 Father with university degree 0.14 0.34 786,673 0.00 0.00 1.00 Father with higher education 0.17 0.38 786,673 0.00 0.00 1.00 Father disability pension 0.01 0.08 786.673 0.00 0.00 0.00	Father income (thousands)	248.85	170.84	765,675	88.67	234.19	402.44
Father with university degree 0.14 0.34 786,673 0.00 0.00 1.00 Father with higher education 0.17 0.38 786,673 0.00 0.00 1.00 Father disability pension 0.01 0.08 786,673 0.00 0.00 0.00	Father western origin	0.89	0.31	765,675	0.00	1.00	1.00
Father with higher education 0.17 0.38 786,673 0.00 0.00 1.00 Father disability pension 0.01 0.08 786.673 0.00 0.00 0.00	Father with university degree	0.14	0.34	786,673	0.00	0.00	1.00
Father disability pension 0.01 0.08 786.673 0.00 0.00 0.00	Father with higher education	0.17	0.38	786,673	0.00	0.00	1.00
	Father disability pension	0.01	0.08	786,673	0.00	0.00	0.00
Father enrolled in edu 0.02 0.15 786,673 0.00 0.00 0.00	Father enrolled in edu	0.02	0.15	786,673	0.00	0.00	0.00
Pregnancy complic., indicator 0.31 0.46 796,416 0.00 0.00 1.00	Pregnancy complic., indicator	0.31	0.46	796,416	0.00	0.00	1.00
N. of schel. CS on day of adm. 0.79 1.34 796,416 0.00 0.00 3.00	N. of schel. CS on day of adm.	0.79	1.34	796,416	0.00	0.00	3.00
Complications at birth 0.57 0.49 796,416 0.00 1.00 1.00	Complications at birth	0.57	0.49	796,416	0.00	1.00	1.00
Emergency CS 0.12 0.33 796,416 0.00 0.00 1.00	Emergency CS	0.12	0.33	796,416	0.00	0.00	1.00
APGAR>7 0.98 0.15 796,416 1.00 1.00 1.00	APGAR>7	0.98	0.15	796,416	1.00	1.00	1.00
APGAR>9 0.91 0.28 796,416 1.00 1.00 1.00	APGAR>9	0.91	0.28	796,416	1.00	1.00	1.00
Post-birth complications, mom 0.05 0.22 796,416 0.00 0.00 0.00	Post-birth complications, mom	0.05	0.22	796,416	0.00	0.00	0.00
Hospital nights at birth 3.47 6.90 791,290 1.00 2.00 5.00	Hospital nights at birth	3.47	6.90	791,290	1.00	2.00	5.00
Discharge on the day of birth 0.36 0.48 796,416 0.00 0.00 1.00	Discharge on the day of birth	0.36	0.48	796,416	0.00	0.00	1.00
Readmitted first 28 days 0.05 0.23 792,754 0.00 0.00 0.00	Readmitted first 28 days	0.05	0.23	792,754	0.00	0.00	0.00
Readmitted first year 0.22 0.42 791,772 0.00 0.00 1.00	Readmitted first year	0.22	0.42	791,772	0.00	0.00	1.00
Readmitted second year 0.16 0.37 791,772 0.00 0.00 1.00	Readmitted second year	0.16	0.37	791,772	0.00	0.00	1.00
Contacts with gp first month 0.70 1.31 792,754 0.00 0.00 2.00	Contacts with gp first month	0.70	1.31	792,754	0.00	0.00	2.00
Contacts with gp first year 13.30 13.01 791,772 1.00 10.00 29.00	Contacts with gp first year	13.30	13.01	791,772	1.00	10.00	29.00
Contacts with gp second year 7.76 9.90 791,772 0.00 5.00 19.00	Contacts with gp second year	7.76	9.90	791,772	0.00	5.00	19.00
Stimulation of labor 0.28 0.45 796,416 0.00 0.00 1.00	Stimulation of labor	0.28	0.45	796,416	0.00	0.00	1.00
Induction 0.19 0.39 796,416 0.00 0.00 1.00	Induction	0.19	0.39	796,416	0.00	0.00	1.00
Neonatal care admission 0.08 0.27 796,416 0.00 0.00 0.00	Neonatal care admission	0.08	0.27	796,416	0.00	0.00	0.00
Nights at neonatal care unit 0.82 5.03 796,416 0.00 0.00 0.00	Nights at neonatal care unit	0.82	5.03	796,416	0.00	0.00	0.00
Death during first year 0.01 0.08 796,416 0.00 0.00 0.00	Death during first year	0.01	0.08	796,416	0.00	0.00	0.00
Perinatal deaths: incl. first 14d 0.00 0.07 796,416 0.00 0.00 0.00	Perinatal deaths: incl. first 14d	0.00	0.07	796,416	0.00	0.00	0.00
Perinatal deaths: incl. first 7d 0.00 0.07 796,416 0.00 0.00 0.00	Perinatal deaths: incl. first 7d	0.00	0.07	796,416	0.00	0.00	0.00
Stillbirth 0.00 0.05 796,416 0.00 0.00 0.00	Stillbirth	0.00	0.05	796,416	0.00	0.00	0.00
Indicator: Epidural 0.24 0.43 192,691 0.00 0.00 1.00	Indicator: Epidural	0.24	0.43	192,691	0.00	0.00	1.00
Indicator: Skin-to-skin 0.76 0.42 192,691 0.00 1.00 1.00	Indicator: Skin-to-skin	0.76	0.42	192,691	0.00	1.00	1.00
Waiting time, epidural (min.) 36.58 53.66 41,421 0.00 25.00 70.00	Waiting time, epidural (min.)	36.58	53.66	41,421	0.00	25.00	70.00
Time from admission to birth (min.) 634.27 954.93 182,524 50.00 330.00 1490.00	Time from admission to birth (min.)	634.27	954.93	182,524	50.00	330.00	1490.00
Time birth to skin-to-skin (min.) 42.92 37.01 129,434 7.10 35.23 95.68	Time birth to skin-to-skin (min.)	42.92	37.01	129,434	7.10	35.23	95.68

Notes: Parental covariates are measured in the calendar year two years prior to child birth. The median, p10 and p90 values are (following Statistic Denmark guidelines) calculated as the average over five observations around the true value. Note that, in our analyses, we set missing values for included covariates to zero and include an indicator variable for missing values for each of the covariates.

Table A.2

Variation in crowding measures in the analysis sample.

	Mean	SD	P25	P50	P75	P90
Range absolute	3.447	2.944	1.000	3.000	5.000	7.000
Range absolute (residualized)	3.047	2.507	1.064	2.276	4.141	6.382
Range percentile	0.328	0.235	0.141	0.289	0.495	0.676
Range percentile (residualized)	0.298	0.214	0.126	0.260	0.440	0.609

Notes: The range in crowding is calculated as follows: For each actual admission we compute the difference between the most crowded and least crowded day, considering the actual day of admission as well as the day before and the day after the actual admission.

Table A.3

The effect of crowding rank on the admission day in the hospital and year on pre-determined characteristics.

	(1)	(2)	(3)	(4)	(5)	(6)	
Mother income	-1.632* (0.874)	0.926 (0.567)	1.186* (0.602)	1.189* (0.603)	1.267* (0.529)	1.346* (0.554)	773,173
Mother uni. deg.	-2.042***	-0.293* (0.172)	-0.129	-0.129	0.081	0.128	796,416
Mother age	-0.066	0.051*	0.064***	0.063***	0.031	0.038*	773,173
Father income	-3.011*	(0.021) -0.445	-0.348	-0.337	-0.340	-0.166	765,675
Father uni. deg.	(1.522) -1.857***	(0.936) -0.060	(0.914) 0.050	(0.914) 0.052	(0.926) 0.230*	(0.922) 0.273*	786,673
Father age	(0.537) -0.079*	(0.138) 0.012	(0.133) 0.024	(0.132) 0.023	(0.126) 0.003	(0.124) 0.009	765.675
Mother CB cont	(0.043)	(0.030)	(0.029)	(0.029)	(0.026)	(0.026)	775 042
	(0.073)	(0.035)	(0.033)	(0.033)	(0.033)	(0.034)	775,545
Mother GP fees	5.771*** (1.174)	3.428*** (0.845)	0.611* (0.329)	0.601* (0.330)	-0.196 (0.363)	-0.172 (0.361)	775,943
Birth weight (gr.)	3.480 (4.188)	-3.024 (3.565)	-3.567 (3.563)	-3.446 (3.571)	-1.908 (3.369)	-1.778 (3.333)	786,530
Hospital FE		Yes	Yes	Yes	Yes	Yes	
Year FE Ouarter of Year FE			Yes	Yes Yes	Yes Yes	Yes Yes	
Day of week FE					Yes	Yes	
Hospital A year I'L						103	

Notes: Each cell presents point estimates from a separate regression. For a description of the main analysis sample, see Section 2.2. The outcome variable for all regressions in a given row is denoted in the first column. Standard errors are clustered at the hospital level. See notes for Table 2 for further details. Significance levels: * p < 0.1; ** p < 0.05; *** p < 0.01.

Table A.4

Robustness: The effect of crowding on transfers and procedure use; short sample (2011-2014).

	Absolute (1)	Percentile (2)	Mean of dep. var	Ν
Admitted to other hosp than first hosp	-0.013 (0.015)	-0.532^{**} (0.245)	18.186	191,746
Stimulation of labor	-0.175*** (0.046)	-2.010*** (0.439)	26.053	192,748
Induction	-0.316*** (0.044)	-3.957*** (0.475)	26.536	192,748
Emergency CS	-0.021 (0.025)	-0.169 (0.329)	12.834	192,748

Notes: See notes for Table 2.

Significance levels: * *p* < 0.1; ** *p* < 0.05; *** *p* < 0.01.

Table A.5

Robustness: The effect of crowding on birth outcomes and child health outcomes; short sample (2011-2014).

	Absolute (1)	Percentile (2)	Mean of dep. var	Ν
Comp. at birth	-0.008	0.073	85.964	192,748
	(0.025)	(0.202)		
Comp. at birth: Laceration degree 1 or 2	-0.069**	-0.985^{**}	45.811	192,748
	(0.032)	(0.439)		
Comp. at birth: Laceration degree 3 or 4	0.016	0.122	3.235	192,748
	(0.011)	(0.144)		
APGAR>7	-0.005	0.023	97.516	192,748
	(0.015)	(0.172)		
APGAR>9	0.014	0.069	91.550	192,748
	(0.032)	(0.337)		
Post-birth complic., mom	0.010	0.028	6.392	192,748
	(0.018)	(0.198)		
Post-birth complic. – severe, mom	-0.009	-0.148	3.927	192,748
	(0.014)	(0.170)		
Hospital nights at birth	-0.002	-0.044	3.023	191,276
	(0.005)	(0.060)		
Neonatal care admission	0.003	-0.171	8.340	192,748
	(0.016)	(0.235)		

J. Maibom, H.H. Sievertsen, M. Simonsen et al.

Table A.5 (Continued)

	Absolute (1)	Percentile (2)	Mean of dep. var	Ν
Readmitted first 28 days	-0.052^{***} (0.016)	-0.575^{***} (0.203)	6.725	191,893
Readmitted first year	-0.046^{*} (0.024)	-0.715** (0.311)	25.512	191,710
Readmitted second year	-0.032 (0.026)	-0.273 (0.344)	16.667	191,710
Contacts with gp first month	0.001 (0.001)	0.008 (0.016)	0.887	191,893
Contacts with gp first year	-0.002 (0.009)	-0.100 (0.088)	8.609	191,710
Contacts with gp second year	-0.009* (0.005)	-0.047 (0.054)	3.148	191,710

Notes: See notes for Table 2 for further details.

Significance levels: * *p* < 0.1; ** *p* < 0.05; *** *p* < 0.01.

Table A.6

The effect of crowding on infant deaths, perinatal deaths and stillbirths (2000-2014).

	Absolute (1)	Percentile (2)	Mean of dep. var	Ν
Death in first year	0.007 (0.005)	0.083 (0.050)	0.583	796,416
Perinatal death (incl. first week)	0.005 (0.004)	0.072* (0.041)	0.464	796,416
Stillbirth	0.000 (0.002)	0.013 (0.023)	0.254	796,416
Mean of 'expected' variation in crowding	3.047	0.298		

Notes: See notes for Table 2. Deaths in first year is an indicator that is equal to one for all stillbirths and children dying during their first year of life. Perinatal deaths include stillbirths and deaths in the first week of life. Stillbirths are defined as deaths on the day of birth conditional on a gestational age of at least 170 days.

Significance levels: * *p* < 0.1; ** *p* < 0.05; *** *p* < 0.01.

Table A.7

Robustness: The effect of crowding on procedures, birth outcomes and child health outcomes, stable wards sample (2000-2014).

	Absolute	Percentile	Mean of dep. var	Ν
	(1)	(2)		
Admitted to other hosp than first hosp	-0.009	-0.381**	11.195	403,960
	(0.016)	(0.172)		
Stimulation of labor	-0.083***	-0.808***	27.023	412,115
	(0.019)	(0.252)		
Induction	-0.228***	-2.331***	18.790	412,115
	(0.030)	(0.273)		
Emergency CS	0.002	0.118	12.159	412,115
	(0.023)	(0.235)		
Comp. at birth	-0.022	-0.072	54.181	412,115
	(0.014)	(0.168)		
Comp. at birth: Laceration degree 1 or 2	-0.082***	-0.791***	35.664	412,115
	(0.020)	(0.263)		
Comp. at birth: Laceration degree 3 or 4	0.017*	0.227**	3.288	412,115
	(0.010)	(0.104)		
APGAR>7	0.002	0.047	97.615	412,115
	(0.010)	(0.096)		
APGAR>9	0.019	0.112	90.998	412,115
	(0.020)	(0.188)		
Post-birth complications, mom	-0.011	-0.074	5.067	412,115
	(0.007)	(0.095)		
Post-birth complications – severe, mom	-0.020***	-0.225**	2.888	412,115
	(0.006)	(0.083)		
Hospital nights at birth	0.003	0.022	3.549	409,417
	(0.005)	(0.059)		
Neonatal care admission	0.028*	0.356**	8.943	412,115
	(0.013)	(0.157)		
Readmitted first 28 days	-0.006	-0.035	5.123	410,194
	(0.011)	(0.114)		

Table A.7 (Continued)

	Absolute (1)	Percentile (2)	Mean of dep. var	Ν
Readmitted first year	-0.001	0.068	21.598	409,708
	(0.029)	(0.279)		
Readmitted second year	-0.029	-0.283	15.757	409,708
	(0.023)	(0.304)		
Contacts with gp first month	-0.000	-0.001	0.688	410,194
	(0.000)	(0.007)		
Contacts with gp first year	-0.013	-0.148	13.624	409,708
	(0.010)	(0.105)		
Contacts with gp second year	-0.011*	-0.095	8.037	409,708
	(0.005)	(0.076)		
Mean of 'expected' variation in crowding	3.063	0.299		

Notes: This table shows results based on analyses using a subsample with only stable maternity wards. A stable ward is defined as a ward that is never experiencing a change in the left vs. right mean of the average monthly number of admissions changing by more than 30 percent. The left mean includes all months up until a given month, the right mean includes months from the current month and onwards (we require at least 6 months of observations before using this rule). The results are similar for alternative cutoffs such as 20 percent or only excluding "unstable" wards only from the month where admissions change above or below the cutoff. See notes for Table 2 for further details.

Significance levels: * *p* < 0.1; ** *p* < 0.05; *** *p* < 0.01.

Table A.8

Robustness: The effect of crowding on procedures, birth outcomes and child health outcomes, alternative crowding measures (moving average or changes in daily admissions, 2000–2014).

	Moving Average (1)	Adm. Change (2)	Mean of dep. var	Ν
Admitted to other hosp than first hosp	0.009	-0.005	18.212	785,336
Stimulation of labor	-0.055*** (0.013)	-0.055^{***} (0.015)	28.055	796,416
Induction	-0.089*** (0.016)	-0.116*** (0.012)	18.612	796,416
Emergency CS	0.004 (0.010)	-0.003 (0.010)	12.022	796,416
Complications at birth	-0.026** (0.012)	-0.019 (0.012)	57.084	796,416
Comp. at birth: Laceration degree 1 or 2	-0.042*** (0.014)	-0.023** (0.011)	34.899	796,416
Comp. at birth: Laceration degree 3 or 4	0.001 (0.006)	0.005 (0.004)	3.164	796,416
APGAR>7	-0.006 (0.005)	0.001 (0.005)	97.642	796,416
APGAR>9	0.007 (0.009)	0.014 (0.012)	91.467	796,416
Post-birth complications, mom	-0.006	0.005	4.889	796,416
Post-birth complications – severe, mom	-0.009** (0.004)	-0.004 (0.003)	2.632	796,416
Hospital nights at birth	-0.001 (0.002)	0.003 (0.002)	3.470	791,290
Neonatal care admission	-0.004 (0.008)	0.010 (0.011)	8.052	796,416
Readmitted first 28 days	-0.004 (0.006)	-0.011 (0.007)	5.409	792,754
Readmitted first year	-0.003 (0.011)	-0.008 (0.011)	22.439	791,772
Readmitted second year	-0.010 (0.012)	-0.007	16.249	791,772
Contacts with gp first month	0.000 (0.001)	0.000	0.700	792,754
Contacts with gp first year	-0.001 (0.003)	-0.008** (0.003)	13.301	791,772
Contacts with gp second year	-0.003	-0.004**	7.756	791,772

Notes: This table shows results where the treatment is defined as the number of admissions of the day of admission including admissions from the previous day ("moving average" specification) or the difference in admissions between these two days (i.e. the daily change in admissions). See notes for Table 2 for further details.

Significance levels: * *p* < 0.1; ** *p* < 0.05; *** *p* < 0.01.

Appendix B. ICD diagnoses and operation codes

To measure birth and post-birth complications, we use diagnoses and operations registered in the Inpatient registry. The codes follow the ICD10 classification. **ICD codes defining "complications at birth"** In our aggregate measure of complications at birth we include diagnoses groups DO60-Do69 and DO71-DO75. To focus on complications that may reflect both severity and the impact of staffhandling, we single out and explicitly focus on a common yet important and closely-monitored complication: Lacerations of degree 1–4 (DO70). Thus we consider separately indicators for lacerations of degree one or two and of degree three or four. To avoid these complications, staffinvolvement in the progression of the birth is vital.

ICD codes defining "post-birth complications" We create two measures of post-birth complications. First, an aggregate measure of post-birth complications. This measure is equal to one for all women having a diagnose of the groups D085-D090 and D0990A. We also create a summary measure of severe post-birth complications that is one if a mother has both a diagnoses from the groups listed above and an operation code in the period from birth to 90 days after birth (operation codes considered here are the groups KMW, KMB, KKCH00, KJFA70, KJFA80, KLCD00, KTAB30).

References

- Aiken, L.H., Clarke, S.P., Sloane, D.M., Sochalski, J., Silber, J.H., 2002. Hospital nurse staffing and patient mortality, nurse burnout, and job dissatisfaction. IAMA 288 (16), 1987–1993.
- Allin, S., Baker, M., Isabelle, M., Stabile, M., 2015. Physician Incentives and the Rise in C-sections: Evidence from Canada. National Bureau of Economic Research.
- Almond, D., Doyle, J.J., 2011. After midnight: a regression discontinuity design in length of postpartum hospital stays. Am. Econ. J.: Econ. Policy 3 (3), 1–34.
- Almond, D., Doyle Jr., J.J., Kowalski, A.E., Williams, H., 2010. Estimating marginal returns to medical care: evidence from at-risk newborns*. Q. J. Econ. 125 (2), 591–634.
- Avdic, D., Lundborg, P., Vikström, J., 2018. Mergers and birth outcomes: evidence from maternity ward closures. IZA Working Paper.
- Bharadwaj, P., Løken, K.V., Neilson, C., 2013. Early life health interventions and academic achievement. Am. Econ. Rev. 103 (August (5)), 1862–1891.
- Chandra, A., Cutler, D., Song, Z., 2011. Who ordered that? The economics of treatment choices in medical care. In: Handbook of Health Economics., pp. 397–432.

- Cook, A., Gaynor, M., Stephens Jr., M., Taylor, L., 2012. The effect of a hospital nurse staffing mandate on patient health outcomes: evidence from California's minimum staffing regulation. J. Health Econ. 31 (2), 340–348.
- Danske Regioner, 2017. Kortlægning Af Svangreomsorgen, Notat EMN-2017-00486.
- Daysal, N.M., Trandafir, M., Van Ewijk, R., 2015. Saving lives at birth: the impact of home births on infant outcomes. Am. Econ. J.: Appl. Econ. 7 (3), 28–50.
- Doyle, J.J., 2011. Returns to local-area health care spending: evidence from health shocks to patients far from home. Am. Econ. J.: Appl. Econ. 3 (3), 221–243.
- Doyle, J.J., Graves, J.A., Gruber, J., Kleiner, S.A., 2015. Measuring returns to hospital care: evidence from ambulance referral patterns. J. Polit. Econ. 123 (1), 170–214.
- Evans, W.N., Kim, B., 2006. Patient outcomes when hospitals experience a surge in admissions. J. Health Econ.
- Facchini, G., 2017. Low Staffing Levels in Maternity Wards: Keep Calm and Call the Surgeon, Working Paper.
- Freedman, S., 2016. Capacity and utilization in health care: the effect of empty beds on neonatal intensive care admission. Am. Econ. J.: Econ. Policy 8 (2), 154–185.
- Harris, M.C., Liu, Y., McCarthy, I., 2019. Capacity Constraints and the Provision of Public Services: The Case of Workers in Public Health Clinics. National Bureau of Economic Research.
- Jensen, V.M., Wüst, M., 2015. Can caesarean section improve child and maternal health? The case of breech babies. J. Health Econ. 39, 289–302.
- Johnson, E.M., Rehavi, M.M., 2016. Physicians treating physicians: information and incentives in childbirth. Am. Econ. J.: Econ. Policy 8 (1), 115–141.
- Macfarlane, A., 1978. Variations in number of births and perinatal mortality by day of week in England and Wales. Br. Med. J. 2 (6153), 1670–1673.
- Marks, M., Choi, M.K., 2019. Baby boomlets and baby health: hospital crowdedness, hospital spending, and infant health. Am. J. Health Econ. 5 (3), 376–406.
- Matsudaira, J.D., 2014. Government regulation and the quality of healthcare evidence from minimum staffing legislation for nursing homes. J. Hum. Resour. 49 (1), 32–72.
- Morley, C., Unwin, M., Peterson, G.M., Stankovich, J., Kinsman, L., 2018. Emergency department crowding: a systematic review of causes, consequences and solutions. PLOS ONE 13 (8).
- Skinner, J., 2011. Causes and consequences of regional variations in health care. In: Handbook of Health Economics., pp. 45–93.
- Snowden, J.M., Kozhimannil, K.B., Muoto, I., Caughey, A.B., McConnell, K.J., 2017. A 'busy day' effect on perinatal complications of delivery on weekends: a retrospective cohort study. BMJ Qual. Saf. 26 (1), e1.
- Sprivulis, P.C., Da Silva, J.A., Jacobs, I.G., Frazer, A.R., Jelinek, G.A., 2006. The association between hospital overcrowding and mortality among patients admitted via Western Australian emergency departments. Med. J. Aust. 184 (5), 208–212.
- Sun, B.C., Hsia, R.Y., Weiss, R.E., Zingmond, D., Liang, L.J., Han, W., McCreath, H., Asch, S.M., 2013. Effect of emergency department crowding on outcomes of admitted patients. Ann. Emerg. Med. 61 (6).
- Woodworth, L., 2020. Swamped: emergency department crowding and patient mortality. J. Health Econ. 70, 102279.
- Woodworth, L., Holmes, J.F., 2019. Just a minute: the effect of emergency department wait time on the cost of care. In: Economic Inquiry.